%&LaTeX
\documentclass{article}

\usepackage[utf8]{inputenc}
\usepackage[T1]{fontenc}
\usepackage{textcomp}

\begin{document}

\section*{2012}

Carles Sanchez, F. Javier Sanchez, Antoni Rosell, \& Debora Gil. (2012).<a target={\textquoteright}_blank{\textquoteright} href={\textquoteright}/files/SSR2012.pdf{\textquoteright}> "An illumination model of the trachea appearance in videobronchoscopy images "</a> In \textit{Image Analysis and Recognition} (Vol. 7325, pp. 313--320). Lecture Notes in Computer Science. Springer Berlin Heidelberg.

\end{document}

